////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-DecimalFormatSymbols]]++DecimalFormatSymbols++::
* ++[[painless-api-reference-DecimalFormatSymbols-getAvailableLocales-0]]static <<painless-api-reference-Locale,Locale>>[] link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getAvailableLocales%2D%2D[getAvailableLocales]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getAvailableLocales%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getInstance-0]]static <<painless-api-reference-DecimalFormatSymbols,DecimalFormatSymbols>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getInstance%2D%2D[getInstance]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getInstance%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getInstance-1]]static <<painless-api-reference-DecimalFormatSymbols,DecimalFormatSymbols>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getInstance%2Djava.util.Locale%2D[getInstance](<<painless-api-reference-Locale,Locale>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getInstance%2Djava.util.Locale%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-DecimalFormatSymbols-0]]link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#DecimalFormatSymbols%2D%2D[DecimalFormatSymbols]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#DecimalFormatSymbols%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-DecimalFormatSymbols-1]]link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#DecimalFormatSymbols%2Djava.util.Locale%2D[DecimalFormatSymbols](<<painless-api-reference-Locale,Locale>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#DecimalFormatSymbols%2Djava.util.Locale%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-clone-0]]def link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#clone%2D%2D[clone]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#clone%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getCurrency-0]]<<painless-api-reference-Currency,Currency>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getCurrency%2D%2D[getCurrency]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getCurrency%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getCurrencySymbol-0]]<<painless-api-reference-String,String>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getCurrencySymbol%2D%2D[getCurrencySymbol]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getCurrencySymbol%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getDecimalSeparator-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getDecimalSeparator%2D%2D[getDecimalSeparator]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getDecimalSeparator%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getDigit-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getDigit%2D%2D[getDigit]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getDigit%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getExponentSeparator-0]]<<painless-api-reference-String,String>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getExponentSeparator%2D%2D[getExponentSeparator]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getExponentSeparator%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getGroupingSeparator-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getGroupingSeparator%2D%2D[getGroupingSeparator]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getGroupingSeparator%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getInfinity-0]]<<painless-api-reference-String,String>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getInfinity%2D%2D[getInfinity]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getInfinity%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getInternationalCurrencySymbol-0]]<<painless-api-reference-String,String>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getInternationalCurrencySymbol%2D%2D[getInternationalCurrencySymbol]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getInternationalCurrencySymbol%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getMinusSign-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getMinusSign%2D%2D[getMinusSign]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getMinusSign%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getMonetaryDecimalSeparator-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getMonetaryDecimalSeparator%2D%2D[getMonetaryDecimalSeparator]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getMonetaryDecimalSeparator%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getNaN-0]]<<painless-api-reference-String,String>> link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getNaN%2D%2D[getNaN]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getNaN%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getPatternSeparator-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getPatternSeparator%2D%2D[getPatternSeparator]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getPatternSeparator%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getPerMill-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getPerMill%2D%2D[getPerMill]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getPerMill%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getPercent-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getPercent%2D%2D[getPercent]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getPercent%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-getZeroDigit-0]]char link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#getZeroDigit%2D%2D[getZeroDigit]()++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#getZeroDigit%2D%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setCurrency-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setCurrency%2Djava.util.Currency%2D[setCurrency](<<painless-api-reference-Currency,Currency>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setCurrency%2Djava.util.Currency%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setCurrencySymbol-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setCurrencySymbol%2Djava.lang.String%2D[setCurrencySymbol](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setCurrencySymbol%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setDecimalSeparator-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setDecimalSeparator%2Dchar%2D[setDecimalSeparator](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setDecimalSeparator%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setDigit-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setDigit%2Dchar%2D[setDigit](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setDigit%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setExponentSeparator-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setExponentSeparator%2Djava.lang.String%2D[setExponentSeparator](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setExponentSeparator%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setGroupingSeparator-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setGroupingSeparator%2Dchar%2D[setGroupingSeparator](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setGroupingSeparator%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setInfinity-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setInfinity%2Djava.lang.String%2D[setInfinity](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setInfinity%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setInternationalCurrencySymbol-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setInternationalCurrencySymbol%2Djava.lang.String%2D[setInternationalCurrencySymbol](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setInternationalCurrencySymbol%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setMinusSign-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setMinusSign%2Dchar%2D[setMinusSign](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setMinusSign%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setMonetaryDecimalSeparator-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setMonetaryDecimalSeparator%2Dchar%2D[setMonetaryDecimalSeparator](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setMonetaryDecimalSeparator%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setNaN-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setNaN%2Djava.lang.String%2D[setNaN](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setNaN%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setPatternSeparator-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setPatternSeparator%2Dchar%2D[setPatternSeparator](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setPatternSeparator%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setPerMill-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setPerMill%2Dchar%2D[setPerMill](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setPerMill%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setPercent-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setPercent%2Dchar%2D[setPercent](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setPercent%2Dchar%2D[java 9])
* ++[[painless-api-reference-DecimalFormatSymbols-setZeroDigit-1]]void link:{java8-javadoc}/java/text/DecimalFormatSymbols.html#setZeroDigit%2Dchar%2D[setZeroDigit](char)++ (link:{java9-javadoc}/java/text/DecimalFormatSymbols.html#setZeroDigit%2Dchar%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Object,Object>>++
